Author's response to reviews

Title: Association of socioeconomic and behavioral factors with adult mortality: Analysis of data from verbal autopsy in Addis Ababa, Ethiopia.

Authors:

   Awoke Misganaw (asterawoke2007@gmail.com)
   Damen Haile Mariam (damen_h@hotmail.com)
   Tekebash Araya (tekebash@gmail.com)

Version: 5 Date: 19 November 2012

Author's response to reviews: see over
Response to a reviewer comment (Andre Pascal A Kengne)

Title: Patterns of mortality in public and private hospitals of Addis Ababa, Ethiopia

Authors would like to thank the reviewer again for the constructive and valuable comments to improve the article further. Please we have given the comments one by one here under with and incorporated in the text.

Discretionary revisions:

1. Presenting fig 1-3 in as a single figure with 3 panels
   Response: We have accepted this comment revised the figures

2. Presenting fig 4 (now numbered 5) to fig 7 as a single figure with 3 panels
   Response: We have accepted this comment revised the figures

3. Presenting fig 8-9 as a single fig with 2 panels
   Response: We have accepted this comment and revised the figures

4. Presenting the curves in figure 10-12 in a single overlaid fig, then showing this new figure together with Fig 13 and 14 as a single figure with 3 panels. The currently superimposed fitted regression lines could be omitted and only the narrative description in the text kept.
   Response: We have accepted this comment and revised the figures and we have excluded the regression line.